Author's response to reviews

Title: Prevalence and predictors of hypertension among adults in Kabul, Afghanistan

Authors:

Islam KM Saeed (km_islam2001@yahoo.com)
Hafez M Rasooly (dochafez@yahoo.com)
Brown JW Nick (nickjwbrown@gmail.com)

Version: 6 Date: 10 April 2014

Author's response to reviews: see over
Cover Letter for revised version

Author's response to editorial request:

Title: Prevalence and predictors for hypertension among adults in Kabul, Afghanistan

Authors:
Dr Khwaja Mir Islam Saeed, (kmislamsaeed@gmail.com) or (km_islam2001@yahoo.com)
Dr Mohammad Hafez Rasooly (dochafez@yahoo.com)
Nick JW Brown (nickjwbrown@gmail.com)

Version: 6 Date: 10 April 2014

Author's response to editorial request:

Dear Editors,

We have gone a long way for this manuscript and finally it came to be reviewed by a native English speaker to be copy edited. Fortunately we were able to access native speaker and experts in copy editing. The style and English have been reviewed by two senior editors of the main UK and European paediatric journal, Archives of Disease in Childhood. I think you ought to be happy with current revised version.

I'm pretty pleased with it and hope it will be accepted for publication now.

Best Regards,

Sincerely,

Khwaja Mir Islam Saeed, MD, MSc

Principle Investigator and Corresponding Author